



Article Combining Genetic Algorithms and SVM for Breast Cancer Diagnosis Using Infrared Thermography

Roger Resmini ^{1,2}^(D), Lincoln Silva ^{2,3}^(D), Adriel S. Araujo ^{2,*}^(D), Petrucio Medeiros ⁴^(D), Débora Muchaluat-Saade ⁴^(D) and Aura Conci ²^(D)

- ¹ Institute of Exact and Natural Sciences, Federal University of Rondonópolis, Cidade Universitária, Rondonópolis 78736-900, MT, Brazil; roger@ufr.edu.br
- ² Visual Lab, Institute of Computing, Fluminense Federal University, Av. Gal. Milton Tavares de Souza, S/N, Niterói 24210-346, RJ, Brazil; lincoln@lampada.uerj.br (L.S.); aconci@ic.uff.br (A.C.)
- ³ Advanced Research Medical Laboratory, Departament of Information Technology and Education in Health, Faculty of Medical Sciences, State University of Rio de Janeiro, R. Professor Manuel de Abreu, 444, Rio de Janeiro 20550-170, RJ, Brazil; petruciomedeiros@id.uff.br (P.M.); debora@midiacom.uff.br (D.M.-S.)
- ⁴ Mídiacom Lab, Institute of Computing, Fluminense Federal University, R. Passo da Pátria 156, Niterói 24210-240, RJ, Brazil
- * Correspondence: adrielsantos@id.uff.br

Abstract: Breast cancer is one of the leading causes of mortality globally, but early diagnosis and treatment can increase the cancer survival rate. In this context, thermography is a suitable approach to help early diagnosis due to the temperature difference between cancerous tissues and healthy neighboring tissues. This work proposes an ensemble method for selecting models and features by combining a Genetic Algorithm (GA) and the Support Vector Machine (SVM) classifier to diagnose breast cancer. Our evaluation demonstrates that the approach presents a significant contribution to the early diagnosis of breast cancer, presenting results with 94.79% Area Under the Receiver Operating Characteristic Curve and 97.18% of Accuracy.

Keywords: breast cancer; diagnosis; thermography; genetic algorithm; support vector machine

1. Introduction

Breast cancer is one of the leading causes of mortality globally, and, last year, it victimized around 685,000 women [1]. Projections indicate that around 12.9 million people will die due to cancer by 2030 and around 857,000 caused by breast cancer, according to the Global Cancer Observatory (GLOBOCAN) [2]. These high mortality rates are associated with late diagnosis of the disease. To improve that perspective, affirmative actions to make the population aware have been employed encouraging prevention and early diagnosis. Early detection and treatment improve such prognosis [3].

Breast cancer diagnosis is carried out mainly by imaging examinations. Mammography highlights the formation of denser tissues (calcifications) and has been widely used worldwide to detect breast cancer. However, this examination is not so efficient for dense breasts and presents substantial false positive rates [4]. Moreover, this acquisition could cause discomfort to the patient and uses ionizing radiation (X-Rays), increasing by up to 2% the patient's chance of developing cancer [5].

Thermography appears as a low-cost alternative, which does not use ionizing radiation, venous access, or any invasive procedure and also can act as a complement to mammography in several scenarios, contributing to the patient's diagnostic process. This examination considers the temperature variation on skin surface [6] to identify the presence of anomalies. However, human interpretation of those images is complex and, for this reason, artificial intelligence algorithms must be used to help the medical team to identify the differences between sick and healthy breast patterns.



Citation: Resmini, R.; Silva, L.; Araujo, A.S.; Medeiros, P.; Muchaluat-Saade, D.; Conci, A. Combining Genetic Algorithms and SVM for Breast Cancer Diagnosis Using Infrared Thermography. *Sensors* **2021**, *21*, 4802. https:// doi.org/10.3390/s21144802

Academic Editor: James F. Rusling

Received: 11 June 2021 Accepted: 5 July 2021 Published: 14 July 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). In this work, we propose a methodology for diagnosing breast cancer based on thermal images. Our proposal builds an ensemble method for selecting models and features by combining a Genetic Algorithm and the Support Vector Machine (SVM) classifier. After a few generations of the Genetic Algorithm, this method selects the models that best fit the problem of breast cancer classification. Subsequently, these models are employed in the features selection process, which seeks to find the smallest features that best classify the data. In addition to generating an effective breast cancer diagnosis method, the combination of these strategies optimizes identifying the parameters and features that best adapt and describe the classification problem. We minimized the most demanding task and maximized fine-tuning choosing the most significant features for breast cancer classification.

Furthermore, the proposed methodology is designed to be applied under different data sources. To assess this methodology's applicability, we evaluate its performance under different perspectives, combinations of databases, segmentation methods, and groups of features. Four experiments are designed, where six sets of features are computed from two different thermographic breast image datasets. In addition, the experiments use different approaches for segmenting the region of interest to generate different data sources.

In the remainder of this paper, Section 2 summarizes some important related work emphasizing their characteristics. Section 3 presents the ensemble and feature reduction approach used in this work. The database, acquisition protocols, and other essential information are presented in Section 4. Section 5 details the core and main contribution of this work: the proposed methodology for the diagnosis of breast cancer. An evaluation of this methodology under different perspectives is presented in Section 6. Section 7 considers experimental analysis and comparison with other work. Finally, in Section 8, conclusions, limitations, and future work are presented.

2. Related Work

The first studies using thermography to detect breast cancer were unsuccessful because infrared camera technology was not able to record temperature variation [7]. Technological advances have allowed cameras to become more sensitive, and temperature differences in the breast infrared thermography of patients with tumors have been highlighted [8]. Posteriorly, Connell et al. [9] indicated that thermal images are suitable to aid in breast cancer detection, evaluation of benign conditions, and follow-up procedures. In this context, thermal imaging has become the focus of several studies involving breast cancer detection in recent decades, as shown by Husaini et al. [10]. In this section, we present some of them.

Arora et al. [11] used 94 thermographic images (60 malignant and 34 benign) from the study at New York-Presbyterian Hospital–Cornell (NPH) to detect breast pathology. Their methodology aimed to find areas of higher temperature difference, evaluating thermal asymmetry between breasts after cooling using the Sentinel BreastScan Software. Their results based on Artificial Neural Networks (ANN) achieved a sensitivity of 97% and specificity of 27%. Wishart et al. [12] evaluated temperature differences and thermal symmetry measures using the same software of the previous study to detect patients with breast cancer. They obtained images from a study conducted in the Cambridge Breast Unit—Addenbrooke's Hospital (CBU). By using artificial intelligence techniques, the authors achieved a sensitivity of 70% and specificity of 48%.

Hossein et al. [13] analyzed 200 thermal images of patients from an experiment that was carried out at Hakim Sabzevari University (HSU) and extracted features of mean, variance, kurtosis, asymmetry, entropy, breast thermal patterns, and difference between breasts and age to diagnose the patient with breast cancer. The authors used a genetic algorithm to select the most significant attributes: difference between breasts, thermal pattern, and kurtosis and obtained a combinatorial model with a 50% sensitivity, 75% specificity, and 70% accuracy.

Krawczyk et al. [14] extracted a series of image features (four basic statistical features, four moment features, eight histogram features, eight cross co-occurrence features, mutual

information, and two Fourier descriptors) from 146 thermal images (29 malignant cases and 117 cases were benign) obtained at Brno University of Technology (BUT). The authors used a Multi Classifier System (MCS) to create multiple random subsets of attributes (ensemble). It is worth highlighting that an ensemble consists of a set of independently trained classifiers [15]. The decision tree-based EG2 algorithm was used for classification. This algorithm uses the information cost function (ICF) to select the attribute of each node to build the tree. The ICF is based on information gain and attribute cost. To select the best MCS classifier, they used a genetic algorithm (GA) with a population fixed at 50 individuals, where each individual represents a set of attributes. The crossover was performed at two points. The selection of the new population was performed using an elitist strategy keeping 10% of the best individuals in the population. The GA stopping criterion was the moment when 100 new populations were generated without performance increase (result stabilization). The ensemble has a number of sets ranging from 1, 3, 5, and 7 and uses 10-fold cross-validation. An accuracy of 91.09%, sensitivity of 80.60%, and

Gerasimova et al. [16] performed a multifractal analysis over a breast temperature time series, in order to detect differences between tissue with malignant tumor and healthy tissue. The authors noticed that complex scalar multifractal properties over autonomic regulation series were drastically altered in cancerous breasts. In this preliminary study, they considered six women diagnosed with breast cancer and three healthy women acquired at Perm Regional Clinical Hospital (PRCH). In another work, Gerasimova et al. [17] considered a larger sample, confirming previous results [16].

specificity of 94.82% were achieved with Ensemble V = 7.

Lashkari et al. [18] proposed an algorithm for cancer classification of thermal images collected from patients with suspected breast cancer referred to Imam Khomeini Hospital (IKH). They extracted 23 features, including statistical, morphological, frequency domain, histogram, and Gray Level Co-occurrence Matrix (GLCM) from segmented right and left breast. They selected the best features by reducing their numbers using the minimum Redundancy and Maximum Relevance, Sequential Forward Selection, Sequential Backward Selection, Sequential Floating Forward Selection, Sequential Floating Backward Selection, and Genetic Algorithm (GA). The authors used as classifiers AdaBoost, SVM, k-Nearest Neighbors (kNN), Naïve Bayes, and probability Neural Network. The best results were obtained for the frontal images with the reduction by GA combined with the AdaBoost classifier, which generated a mean accuracy of 85% and 87% for the left and right breast.

Raghavendra et al. [19] presented a study that allows diagnosing patients using a descriptor based on a histogram of oriented gradients (HOG) with kernel locality preserving projection (KLPP) for feature reduction. The authors used 50 thermal images (25 healthy and 25 with malignant tumor) acquired at the Department of Diagnostic Radiology—Singapore General hospital (SGH). This methodology used Decision Tree, Discriminant, k-Nearest Neighbor and Fuzzy Sugeno, Naive Bayes, SVM, AdaBoost, and Probabilistic Neural Network classifiers. The results indicated that the Decision Tree classifier reached an average accuracy, sensitivity, specificity, and area under curve of 98%, 96.66%, 100%, and 0.98%, respectively.

Santana et al. [20] studied the classification of breast lesions (cysts and benign and malignant lesions) from thermal imaging using Extreme Learning Machine (ELM). They extracted features of texture and shape with moments of Haralick and Zernike and used the features separately and combined. Their results with ELM reached 95.56% using the combination of the features. Posteriorly, Santana et al. [21] analyzed the performance of several classifiers and concluded that ELM presented the best result with accuracy of 73%, sensitivity of 78%, and specificity of 88%.

Based on the study of Santana et al. [20,21], Silva et al. [22] used texture and shape features computed as moments of Haralick and Zernike. The authors reduced the amount of features using GA and Particle Swarm Optimization (PSO). Their result was classified by: Bayes Net, Naïve Bayes, Multilayer Perceptron, SVM, J48, Random Tree, Random Forest, and Extreme Learning Machine (ELM). The result demonstrated that AG presents a

slightly better performance compared to PSO. The ELM classifier with a polynomial kernel of exponent 3 achieved better performance with accuracy equal to 88%.

Baffa and Lattari [23], Fernández-Ovies et al. [24], and Tello-Mijares et al. [25] use a Convolutional Neural Network to classify patients into healthy and unhealthy using thermal images from the DMR-IR database [26]. Baffa and Lattari [23] used DIT (for final diagnosis) and SIT (for initial screening of cases) and obtained 98% of average accuracy with DIT images and 95% with SIT images. Fernández-Ovies et al. [24] obtained best performance with Restnet34 and Resnet50, reaching a prediction of 100%. Tello-Mijares et al. [25] compared the performance of CNN with other classifiers (random forest, multilayer perceptron, and Bayes network): CNN achieved 100% accuracy.

Silva et al. [27] used Support-Vector Machine (SVM) to classify patients as healthy or with breast abnormalities. Their first step is the definition of a Region of Interest (ROI) of each sequence of images. From the ROI, a temperature time series was built, and several features were calculated. The main features were automatically selected in the WEKA (Waikato Environment for Knowledge Analysis) tool using the CfsSubsetEva attribute evaluator and the BestFirst research method and then classified by SVM. The authors achieved 100% accuracy and used the DMR-IR database [26].

Sánchez-Ruiz et al. [28] proposed a methodology for classifying the breast as normal and abnormal. The main contribution of their work was the development of an automatic segmentation of ROI. After the segmentation step, they extracted first and second order statistical features by computing histogram and GLCM for each ROI. They used a GA to reduce the number of features and increase the classifier's performance. The results obtained were: 98% accuracy, 97% sensitivity, and 100% specificity using the Artificial Neural Networks (ANN) classifier.

Mishra and Rath [29] extracted the Gray level Run Length Matrix (GLRLM) and GLCM features to classify patients between healthy and unhealthy. These features were selected using Principal Component Analysis and Autoencoder and applied to SVM, decision tree, random forest, K-NN, linear regression, and fuzzy logic classifiers. Their results demonstrate that random forest with PCA achieved the best performance with an accuracy of 95%.

Table 1 presents a summary of the related work indicating: the year, general goal (screening and diagnosis), image acquisition protocol (SIT and DIT), number of samples in each class: healthy (H), cysts (C), benign tumor (B) and malignant tumor (M), and database used.

Authors	Voar	Sara	Diag	דום	DIT SIT		Number of Samples			Database
Autions	Ieal	Scie.	Diag.	DII	511	Н	С	В	М	Database
Arora et al. [11]	2018	Х	Х	Х	-		34		60	NPH
Wishart et al. [12]	2010	Х	Х	Х	-		41		65	CBU
Hossein et al. [13]	2012	-	Х	-	Х	185	-	-	15	HSU
Krawczyk et al. [14]	2012	-	Х	-	Х	-	-	117	29	BTU
Gerasimova et al. [16]	2014	-	Х	Х	-	3	-	-	6	PRCH
Gerasimova et al. [17]	2014	-	Х	Х	-	14	-	-	33	PRCH
Lashkari at al [19] *	2016		v	Х	-	46		21		IKH (Left Breast)
Lashkari et al. [10]		-	л			34		33		IKH (Right Breast)
Raghavendra et al. [19]	2016	-	Х	-	Х	25	-	-	25	SGH
Santana et al. [20]	2017	-	Х	-	Х	-	73	121	76	UFPE
Santana et al. [21]	2018	-	Х	-	Х	-	219	371	235	UFPE
Baffa and Lattari [23]	2018	Х	-	-	Х	177		42		DMR-IR
Dalla alla Lattali [23]	2010	-	Х	Х	-	95	-	-	42	Divin-in
Fernández-Ovies et al. [24]	2019	Х	-	-	Х	175		41		DMR-IR
Tello-Mijares et al. [25]	2019	Х	-	-	Х	35		28		DMR-IR
Sánchez-Ruiz et al. [28]	2020	Х	-	-	Х	142		33		DMR-IR
Mishra and Rath [29]	2020	Х	-	Х	-	19		37		DMR-IR
Silva et al. [27]	2020	Х	-	Х	-	32		32		DMR-IR
Silva et al. [22]	2021	-	Х	-	Х	66	73	121	76	UFPE

Table 1. Comparison of related work using thermographic images.

* Number of samples from the right (first line) and left (second line) breasts.

3. Ensemble and Dimensionality Reduction

Ensemble methods are an important approach in machine learning techniques, and can be presented in the literature under different names, such as committees of learners, mixtures of experts, classifier ensembles, multiple classifier systems, consensus theory, etc. [30]. The main idea behind the ensemble methodology is to construct a set of models by combining them to solve the problem, and, in a traditional approach, a single model is built around the data [31]. According to [32], there are four levels to build an ensemble, varying according to what is manipulated to build the system:

- 1. Data level: manipulation is done on attribute values. Several sets of instances are created according to the data value and submitted to the estimator;
- 2. Feature level: manipulation is done on instance attributes. Different sets of features are created and submitted to the estimator;
- 3. Classifier level: manipulation of classifiers. Different base classifiers are used to compose a final estimator;
- 4. Model level: the manipulation is performed on the models, selecting which ones will compose the final estimator. Different models that may or may not come from different base classifiers are combined to result in a single estimator, combining the results of each model.

The strategy of manipulating the learning algorithm is also discussed by the authors of [33], who state that, in this technique, the same base classifier can be configured in different ways to classify the same set of instances, thus generating a variety of classification models. The stacking algorithm is an example of this category.

The area of combining classifiers can be further divided into two possible strategies: classifier fusion systems or classifier selecting systems [34]. The result is the final model by a fusion of better options or selecting from among the options. The fusion or selection process represents the last step of the system. In a problem where the end result is a model that stands out over the other models, the best solution is probably a fusion of better models, while selection is more advisable for problems where the winning model has an overall score, which may be by majority vote, significantly superior to that of its competitors.

In the fusion process, it is complicated to bring together the decisions of several models, especially if they are from different base classifiers, and the problem with the selection is that there must be an "oracle" capable of correctly asserting which model is the winner of the group [32]. In this work, we focus on the selection approach. The flow to deal with ensembles using the selection strategy can be divided into two phases: the generation of the set of models and the selection itself.

In generation phase, an important task involves generating a set of mutually complementary classifiers whose combination provides optimal performance. Furthermore, it can be argued that a success set is one whose generated predictors are accurate and which makes mistakes in different parts of the input space. Another important point is that a greater diversity of models positively impacts the system's results. When models are generated using the same induction algorithm, the set is called homogeneous; otherwise, it is called heterogeneous.

Different models can be experts at different points in the search space (great places for the general problem). Therefore, an appropriate approach is to select the most suitable model or set of models from the group. This approach is known as Dynamic Classifier Selection (DCS), when it selects a single model, or Dynamic Ensemble Selection (DES), when a selection outcomes a set of models (an ensemble). However, the selection of models can be static or dynamic. In the static selection, the selection of models is performed in the training phase and, in the dynamic selection, the selection has been a preferable solution.

Whichever strategy is chosen, the biggest problem in model selection is in the evaluation strategy of the best model or set to be selected. A usual approach is to use the result of the model in the test phase to compose a ranking and make the choice from this list. Bucket of models is an ensemble technique in which a model selection algorithm is used to choose the best model from the set. Therefore, a "bucket of models" approach cannot perform better than the best model in the set [35].

A well-known ensemble learning method is stacking. It is a generic framework that combines several ensemble methods and an enhanced extension of bucket of models that supports heterogeneous models in the generation of ensembles [36]. Stacking has two levels of learning: basic learning and meta-learning. In the first one, models are trained with the training dataset. Once trained, models create a new dataset for a meta-model. The meta-model is then trained with this new training dataset. Finally, the trained meta-model is used to classify new instances.

However, defining the best model or set of models is a costly process, especially if the dataset contains many columns (attributes). In addition, many have a strong dependency between the dataset and the defined model. Selecting a subset of attributes with a very small number of samples and a large number of attributes represents a serious challenge. The success of an attempt to select attributes from a large dataset depends on several factors: the underlying probability distributions (some problems can be easy to solve), the sample size (number of instances), the dimensionality (number of attributes), the method chosen for attribute selection (how well it finds good subsets of attributes, how robust it is for overfitting, how precisely the criterion of interest is evaluated), and the classifier subsequently recommended to the user. Recent resource selection approaches consider evolutionary learning as one possible way to deal with them [37].

4. Datasets

We used the thermal images available at the Database for Mastology Research with Infrared Image (DMR-IR) (The Database for Mastology Research with Infrared Image (DMR-IR) is available at http://visual.ic.uff.br/dmi, accessed on 29 June 2021) and the private thermal image database of the Federal University of Pernambuco (UFPE). Thermal images are acquired through a camera sensitive to infrared radiation that varies according to the equipment used. The DMR-IR uses a FLIR SC620 and the UFPE database uses a FLIR S45 camera.

DMR-IR database was approved by the Research Ethical Committee, registered under the number CAAE: 01042812.0.0000.5243 at the Brazilian Ministry of Health platform (Brazilian Ministry of Health platform is available at http://plataformabrasil.saude. gov.br, accessed on 29 June 2021). UFPE database also was registered in the Brazilian Health Ministry and approved by the Ethics Committee of University of Pernambuco (CEP/CCS/UFPE n° 279/05).

4.1. DMR-IR

DMR-IR manages and retrieves breast exam information and clinical data from volunteers. Its goal is to support the scientific community in the development and comparison of computational methodologies that assist in the detection and diagnosis of breast cancer. It has SIT and DIT images, but in this work we use only SIT images.

In the static protocol defined in DMR-IR, five (5) thermal images (one frontal and two lateral left and right in the positions of 45° and 90°) are acquired from volunteers standing with hands on their head, after acclimatization for 10 min in a room with the controlled temperature between 20 °C and 22 °C (see Figure 1).



Figure 1. Thermogram acquisition from volunteer [38].

From this database, we selected 80 examinations with 40 "no cancer" and 40 "cancer" diagnosis. The group "no cancer" contains normal and benign cases.

4.2. UFPE

The protocol established for the acquisition of thermal images from UFPE uses equipment that allows the volunteer to sit and uphold their arms. At the beginning of the image acquisition protocol, the temperature of the room is recorded and thermal images are acquired after acclimatization for 10 min. The images are acquired from the frontal position with hands on the head and waist, with a metallic grid, internal side of the left and right breasts, and external side of the left and right breasts.

Unlike DMR-IR, the thermal images in this dataset are not publicly available. We selected 98 examinations with 60 "normal" and 38 "cancer" from the UFPE database. An interesting point to note is that the distribution of cases from the UFPE database makes it unbalanced. To deal with unbalanced classes, we used performance evaluation measures that do not privilege the majority class, such as Area Under the Roc Curve (AUC), F1 Score(F1), sensitivity (SENS), and specificity (SPEC) [39]. The goal is to show that the proposed methodology can provide the best classifiers for both, unbalanced and balanced bases.

5. Proposed Methodology

This work presents a methodology to diagnose breast cancer based on thermal images using an ensemble strategy. The ensemble strategy proposed in this work is based on a model selection strategy using SVM as a single base classifier. As the oracle of the selection system, we use a ranking of the performance of each model in a dynamic ensemble selection using a Genetic Algorithm capable of generating and selecting the most adapted models. We also built another Genetic Algorithm to perform the attribute selection evaluated by the models resulting from the previous Genetic Algorithm.

The main idea is based on two ensembles, one for model selection and one for attribute selection. The first ensemble aims to select models partially adapted to problems in a few generations of the Genetic Algorithm. This strategy aims to save resources, as, at this stage, all dataset attributes are sent for model training and testing. A set of best models is selected at the end of the process (the best models adapted in a few generations). The next ensemble aims to select the smallest set of attributes that best fit the data. As a fitness function in this step, the set of models selected in the previous step is used. We use another Genetic Algorithm for this step, and this time with more generations looking to achieve the best result by finding the best feature set that generates the best classification result without changing the base classifier configuration. We take into account the dependency between classifier and attributes [37], minimizing the heaviest task and maximizing the fine-tuning that is choosing the most significant attributes for the problem.

The proposed system presents three stages. In the first, there is the selection of models, performed by the Bucket of Models technique. The outcome of the Bucket of Models is a set of ten models that do the classification of the problem but are selected after a few generations of the Genetic Algorithm. This set is then taken to the second stage, where another Genetic Algorithm performs the feature selection, testing several sets of features. The outcome of the second Genetic Algorithm is the smaller set of features that best describes the problem. At the end of this process, the resulting model is used to classify the patient (diagnosis). Figure 2 illustrates the general flow presented in this work.



Figure 2. General flow of the proposed methodology.

We choose a Genetic Algorithm to perform Bucket of Models and the feature selection because the Genetic Algorithm is indicated to be used in scenarios where there is not much data available for evaluation (the amount of breast thermal images is not huge). In addition, the theoretical analysis (knowledge base) of the problem is also not extremely rich (the use of thermal images is not a consensus among physicians and the analysis is performed visually many times). Thus, we can affirm that data and theory about this problem are poor. Another fact to support this scenario is that the golden standard exam to classify breast cancer is mammography that shows no more than 80% of accuracy in a visual analysis. In other words, the analysis is subjective.

Using these techniques is in line with the Theory of Intelligent Systems [40], illustrated in Figure 3, where the ideal scenario of applying the Genetic Algorithm is positioned in a middle path where there is not much data available and the theory on the problem is not notably high. Alternatives such as Convolutional Neural Networks and Deep Neural Networks, on the other hand, could perform better when working with a larger amount of data.



Figure 3. Usability of different methods for problem-solving depending on availability of data and expertise (theories) on a problem [40].

The proposed methodology is designed to be applied and functional for different particularities of the databases, providing the set of models and features best suited to diagnosing breast cancer. In this work, we evaluated this methodology (Section 6) using two distinct databases that have thermographic breast exams captured under different acquisition protocols. Furthermore, for each database, we investigated the methodology's performance working under different groups of features computed from distinct regions of interest.

The particularities of the model and feature selection process, as well as the use of the Bucket of Models, Genetic Algorithms, and SVM, are detailed in the next subsections.

5.1. Model Selection

Machine learning algorithms are used, among other tasks, to build a model for classification. The construction of this model involves the choice of a basic classifier, which, in turn, can admit a series of parameters that assume different values. Each change in these values results in a new model that fits the data differently and presents different performance values. Thus, defining the best parameters that will fit a model is a search problem.

The most widely used approach to model selection is cross-validation selection, which can be described as "try all models and choose the one that produces the best result" [41]. This approach, however, requires an exhaustive search among all the options of model configurations. Several times, the researcher performs these configurations manually varying such parameters [42]. The choice of the best configuration of a classifier can be made by combining (by selection or fusion) classifiers in several ways. In this sense, we have the techniques of Bagging, Boosting, Staking, and Bucket of Models.

Bucket of Models

The Bucket of Models is responsible for providing all possible model configurations from: (a) a base classifier and (b) the set of parameters for such a base classifier, as well as their respective values. Differently from other techniques, the Bucket of Models selects the best fit model from several models (bucket).

We use a Genetic Algorithm to manage the Bucket of Models. A Genetic Algorithm works generating successive new populations that are a set of individuals, and each individual represents a model. Thus, a Genetic Algorithm provides a Final Set, which in this stage refers to the set of models that best adapt to the classification problem. Figure 4 illustrates the process adopted by the Bucket of Models in our study. Combining the Base Classifier with the Set of parameters will generate the Possible Models to be used in the Evaluation process through the Genetic Algorithm. To evaluate these models, the Genetic Algorithm will also use the set of features that describes the classification problem as input Data. These data are used for training and performance evaluation of each generation of the Genetic Algorithm models. The models resulting from this process will be used in the subsequent step of this methodology (feature selection).



Figure 4. Model selection general flow.

5.2. Feature Selection

The outcome from the first step is a solution set with models best adapted to the classification problem. However, these models may not necessarily present the best results. This is the equivalent of stating that the models are great candidates to solve the classification problem.

Therefore, selection of features improves the final result of the classification by evaluating a set of features that best describes the problem. This task maximizes the correctness of the classification since the reduction in the number of features can improve the classification results [33].

For this task, we use a Genetic Algorithm where each individual is composed of a sequence of features computed from the images. For each individual, some features are enabled, and others are not. The Genetic Algorithm evaluates, through successive new generations, these individuals and provides the smaller set of features (the Final Set) that best describes the classification problem. For the Evaluation process, in each generation, the Genetic Algorithm for feature selection will also use the set of features that describes the classification problem as input Data. At the end of this process, the ensemble proposed in this study will select the Best Features, the smaller set of features with the best outcome. As many problems could be unbalanced, we define Area Under the ROC Curve as a performance measure. Figure 5 illustrates the flow of feature selection. Figure 5 illustrates the flow of feature selection.



Figure 5. Feature selection general flow.

Unlike grid search, the proposed method does not waste time searching in nonpromising regions, saving processing resources and computation time. In addition, the algorithm inserts certain randomness into the exploration process, making some searches even in unexplored regions.

5.3. Genetic Algorithm

A Genetic Algorithm is an evolutionary algorithm that generates solutions of search problems using techniques inspired by the natural evolution of species [43]. Solutions based on genetic algorithms usually generate several populations of individuals based on evolutionary strategies, such as heredity, mutation, and natural selection [44]. In this work, we used a Genetic Algorithm that combines all these techniques.

As we mentioned before, there are several strategies to evaluate the result of ensembles. We choose to use the classifier itself as the performance evaluator. Thus, the Fitness Function is composed of a model created from the base classifier.

In addition to this proposal of using GA, this study also considers the occurrence of *decimation events*. When these events happen, few individuals survive and the rest are randomly created. In this proposal, such events happen if the solution set has presented the same performance for several generations, and we call this *Perturbation*. Figure 6 illustrates the general flow of the genetic algorithms proposed in this work.



Figure 6. Genetic Algorithm flow.

In the Genetic Algorithm flow, the first population of individuals is randomly generated ("Generate Initial Population" step) and subsequently taken for evaluation. For the Evaluate new population step, the objective function will evaluate each individual in the population, based on a performance measure, and create a Ranking list. If the Stopping criterion is not met, a new population is generated at the end of the evaluation.

If the Stopping criterion is met after the population has been evaluated, the algorithm ends the cycle of creating new generations and Selects the best fit for the Final Set. In the model selection, this set corresponds to the best models, and, in the feature selection, it presents the model with the best set of features in terms of performance.

5.3.1. Model Selection Version

Genetically speaking, what makes an individual different from its ancestors are the evolution perceived in its DNA, composed of a sequence of genes. For the model selection of this work, DNA can be understood as the sequence of parameters that each model assumes. Each of these parameters represents a gene of the individual. Thus, with each new generation, evolution takes place in individuals modifying their genes.

In the Crossover stage, genes are modified following the concept of heredity, where the genetic load of the new individual is formed from their parents' genes. In the approach proposed in this work, each descendant receives, at random, genes from both parents. The genetic load is configured to be balanced, receiving the same number of genes from each parent in case of even numbers of them—or almost a similar number of parameters in case of odd numbers of elements. Figure 7 illustrates the crossover process for the genetic algorithm used for the model selection. The number of individuals generated by crossover, in each generation, can be configured by the user.



Figure 7. Crossover reproduction in a model selection step.

In asexual reproduction, an individual's DNA is altered. Thus, this change makes him a new individual in conceptual terms. In this work, this reproduction method refers to the alteration of the values of each model's parameters (genes). Assuming that a gene X has the value *n*, the reproduction will generate two new individuals, where, in one of them, gene X has the value n - 1, and, in another, it produces the value n + 1. This process is repeated for all genes, and represents a fine-tuning of some individuals in the population (the most adapted in the current population) to improve individual performance. In the task of manually adjusting the estimator parameters, the researcher sometimes updates the model with small variations in the parameters. This step tested in our study simulates this behavior. Figure 8 illustrates the flow of the evolution of individuals based on asexual reproduction. In this work, the number of new individuals generated by each of these approaches can also be configured by the user.



Figure 8. Asexual reproduction for the genetic algorithm.

Conceptually, the evolution of individuals can also occur by mutation. In the implemented model selection, in each generation, ten individuals undergo mutation: the mutation changes only one gene of the individual to any other acceptable value. However, in each new generation, ten mutations occur, one among the three best, one among the three worst and the other eight occurring randomly in the entire population. The mutation rate is high because it is intended to force the population to high rates of renewal. Unlike creating an individual at random, the mutation works on only one individual's gene.

Moreover, a new generation may contain randomly generated individuals without necessarily having a genitor. In these cases, in practical terms, the parameters of each new model receive random values (considering the allowed values for each parameter). This randomness helps the solution seek different results that may eventually perform better than the current ones. Consequently, part of the individuals of each generation are formed randomly.

In addition to the reproduction methods, it is also important to mention that individuals considered stronger are selected to compose the new generation. This process is similar to real life, where, in each generation, we have members from previous generations who remain vital for society. This process is only interrupted if a decimation event occurs, where only the strongest individual resists, and all other population members will be randomly generated.

5.3.2. Feature Selection Version

The genetic algorithm mechanisms used for feature selection follow premises similar to those previously presented. In each iteration, the strongest members of the population also make up the new generation. In addition, evolution techniques are also considered, modifying the genetic load of another part of individuals. However, as the purpose of this step is to find the set of features that best describes the problem, each individual's genes now refer to the features themselves. Unlike the previous step, where each gene could assume different values, each gene is binary, assuming only two states: active or inactive. The active genes represent the set of features that the individual is evaluating. The set of active features changes for each individual. Figure 9 illustrates the composition of each individual in the feature selection step.



Figure 9. Individual composition in the feature selection step.

Sexual reproduction also occurs from the crossing of two individuals, where each contributes with a genetic part to the formation of their descendants. Each pair of genitors generates two descendants. The number of genes from each genitor is randomly defined, obeying a minimum value (defined by the user). Figure 10 illustrates an example of this process, where the individual has 18 genes. In this example, the cutoff point is 6 (chosen at random), so the descendants of this couple will be composed of the following genetic load: (a) Descendant 1: genitor 1 will contribute to the first six genes in his load; the rest will come from the last genes of genitor 2; (b) Descendant 2: genitor 2 will contribute to the first six genes in his load; the rest will come from the last genes of genitor 1.



Figure 10. Crossover reproduction in the feature selection step.

As the genes, in this stage of the work, assume binary values, the total number of different potential individuals (considering the complete execution of the genetic algorithm) is smaller. For this reason, the form of asexual reproduction is considered only by mutation (which occurs in a low percentage of the population). As in the previous step, the mutation changes only one gene of the individual (activating or deactivating it). The mutations are programmed to happen at least once in: (a) some of the strongest members (set of features with the best performance); (b) some of the weakest members (set of features with the lowest performance); and (c) any other member of the population.

Random reproduction selects which genes will be activated for each new individual. Devastating events are also considered, where only the strongest member survives, and the rest are randomly generated.

5.4. Patient Classification

On completing the model and features selection steps, the ensemble proposed in this work delivers a final model for the classification of breast cancer diagnoses. This model combines the parameters that best adapt to the problem and the best set of features for classifying the images.

6. Proposal Evaluation

To consolidate and evaluate all the new aspects proposed in this paper, an evaluation to observe the behavior of the proposed ensembles under the influence of different data inputs was done. For this, different groups of images were considered and, for each of these groups, features based on different computational techniques were computed and applied. Each variation in the input data configures different aspects to evaluate the methodology proposed in classifying breast cancer diagnoses.

6.1. Considerations Related to the Databases and the Segmentation Aspect of the Input Images

Two bases of thermal breast images were selected to compose the scope of our evaluation. For each of these bases, two segmentation approaches for obtaining the ROI were used: (a) one in which the evaluated region consists only of the breast area, without considering adjacencies [22,45]; and (b) another one in which the evaluated region extends between the breast and part of the armpits area [20]. This second ROI type aims to consider the lymph node region responsible for transporting the lymph from the mammary tissues to the circulatory system [3]. The combination of two databases and two segmentation methods results in four different configurations of experiments.

For each of these experiments, several sets of features are computed [38]. Each set aims to evaluate, under different data inputs, the behavior of the system under the combinations of classifiers considered in our methodology. Four different approaches to compute features are considered:

- 1. Gray Level Co-occurrence Matrix (GLCM) [46]:
- 2. Local Ternary Pattern [47];
- 3. Daubechies Wavelet [48];
- 4. Higuchi [49], Petrosian Fractal [50] Dimensions and Hurst Coefficient [51].

In our evaluation, we combined all approaches into six different groups of features. This diversification in several groups allows us to assess the adaptability of the proposed methodology in classifying diagnosis from thermal images based on different features. Table 2 illustrates these groups and the total number of features formed in each one.

Table 2. Details related to the description of leatures.	Table 2. Details related to the	e description of features.
-----------------------------------------------------------------	---------------------------------	----------------------------

Description of the Group	N. Features	Detail of the Features
Fractais + Wavelets + 8 LTP	384	3 Fractals \times 8 Wavelets \times 8 LTP \times 2 Breasts
Fractais + 8 LTP	48	3 Fractals $ imes$ 8 LTP $ imes$ 2 Breasts
GLCM	48	6 Descriptors \times 4 Directions \times 2 Breasts
Fractais + 2 LTP + GLCM	60	3 Fractals \times 2 LTP Spectrum (0 and 4) \times 2 Breasts + GLCM
Fractais + 3 LTP	18	3 Fractals \times 3 LTP Spectum (0, 3 and 6) \times 2 Breasts
GLCM + 8 LTP	148	100 LTP features + GLCM

6.2. Considerations Related to the Input Parameters

We use the SVM classifier [52] as a base classifier for our evaluation because several studies applied to breast cancer for screening or diagnosis have considered this the most adequate classifier for their solution as well [53,54]. This classifier assumes a number of parameters that can be adjusted in order to achieve better performance. In this work, we

explore parameters Tolerance, Gamma, Coef0, Nu, Degree, and Kernel types [52], with the following values:

- 1. Tolerance: [0.1, 0.01, 0.001, 0.0001, 0.00001, 0.000001, 0.0000001];
- 3. Nu: [0.2, 0.3, 0.4, 0.5, 0.6];
- 4. Degree: [1, 2, 3];
- 5. Kernel: ['rbf', 'poly', 'sigmoid'].

The various combinations between these values will be registered by the Bucket of Models to be used by the Genetic Algorithm to select the models that best adapt to the classification problem.

The ensembles introduced in this work intend to generate good results for diagnosing breast cancer, training a substantially smaller number of models than the search grid, which, as previously mentioned, is considered the most widely used approach for model selection [41]. Thus, the selected range to be explored by the Genetic Algorithm and SVM in the model selection step assume the variations that a search grid would usually practice, varying the values according to the specificities of each parameter [55].

6.3. Considerations Related to Reproduction Surroundings

The proposed ensembles allow the user to configure the number of individuals for each approach to population evolution and the total number of generations. In our evaluation, the selection of models will run under 30 generations where, in each generation, the total population will be 140 individuals. The version of the genetic algorithm used for the feature selection will also work with 140 individuals, testing another set of configurations for the evolution of the individuals under 100 generations. The selected number of individuals generated by each reproduction strategy is presented in Table 3.

	Individuals by		
	Model Selection	Feature Selection	
Selection	10	20	
Crossover	40	40	
Asexual reproduction	20	-	
Mutation	10	6	
Randon	60	74	

Table 3. Individuals generated by each reproduction strategy.

These configurations are bioinspired. For instance, crossover reproductions tend to be one of the most common forms of reproduction for animals. For this reason, a significant share of the elements of each generation is generated by crossover. In addition, asexual reproductions also occur in different ecosystems, especially those botanic related, thus it also composes the populations of this study. Mutations tend to happen in both fauna and flora, but in a smaller percentage, reflecting on the number of mutations applied in this study. Finally, for vegetables and animals, characteristics from past generations also tend to compose new generations, inspiring the configurations of the proposed method.

6.4. Experiment Design

We ran four experiments to evaluate our proposal:

• Experiment 1: images from the DMR-IR database, and as ROIs the area of breast and armpits as presented in Figure 11b.

- Experiment 2: images from the DMR-IR database, and as ROIs only the breast area, without adjacencies as presented in Figure 11c.
- Experiment 3: images from UFPE database, and as the ROIs the breast with armpits as presented in Figure 11e.
- Experiment 4: images from UFPE database, and as ROIs only the breast area, without adjacent parts as presented in Figure 11f.

For each experiment, the evaluation of every new model, generated in the model selection and the feature selection steps, is performed by the Genetic Algorithm in the Evaluation step. In this phase, each model is trained considering 70% of the data instances and tested, considering the remaining 30% instances, both using the *k*-*fold* cross-validation where k = 4. The details of each database are described in Section 4. For each experiment, all features described in Section 6.1 are computed. Table 4 summarizes these experiments.



Figure 11. Samples of images from the used datasets, their default false colors and ROIs: (**a**) Image from DMR-IR dataset; (**b**) ROI based on the breast and linfonodes area using DMR-IR images; (**c**) ROI based only on the breast area using DMR-IR images; (**d**) Image from UFPE dataset; (**e**) ROI based on the breast and linfonodes using from UFPE images; (**f**) ROI based only on the breast area using UFPE images.

Table 4. Main elements considered in the experiments.

Experiment	Database	ROI	Features
#1	DMR-IR	Breast + armpits	
#2	DMR-IR	Breast	All fostures presented in Table 2
#3	UFPE	Breast + armpits	All leatures presented in Table 2
#4	UFPE	Breast	

7. Results and Discussion

The Python programming language, and its libraries and frameworks were used. The feature computation process used methods implemented for this work using the Scikit-Image library, which has a collection of algorithms for analysis and image processing [56]. The model training and testing activities used components from the Scikit-Learn framework, which presents algorithms for machine learning activities in Python [55].

To validate the proposed methodology, the four experiments mentioned in Section 6.4 combining information computed from different areas (ROIs with and without armpits) and using images from different databases, features, and classification approaches were performed as a binary problem, where the instances of features may only have one of two classes: cancer and non-cancer.

7.1. Experiment 1

The execution of this experiment had as input the features computed from the thermograms extracted from the DMR-IR. These thermograms were segmented considering the corresponding area between the breast and the armpits as ROI.

Table 5 summarizes the results obtained in this experiment, using AUC as a measure of performance. For each of the groups of features used in this experiment, the performance is presented in two moments: intermediate and final. The intermediate moment represents the best result obtained after completing the model selection. In this step, the genetic algorithm was performed on the various parameter configurations and defined those that best adapt to the classification problem. This result, however, still does not consider the execution of the Genetic Algorithm for feature selection, a crucial step in the proposed methodology. After selecting features, we can see that the final results of the methodology proposed in this study are, for all combinations of features, greater than 80%, reaching up to 96.15% for some groups of features.

These results indicate that the proposed methodology is effective for any group of features considered. In this way, it is possible to observe that this solution seems to be generic enough to deal with different features and presents promising results.

|--|

	AUC (%)				
Group of Features	Intermediary	Final			
	(Model Selection)	(Feature Selection)			
Fractals + Wavelets + 8 LTP	80.12	94.87			
Fractals + 8LTP	80.12	96.15			
GLCM	80.12	94.87			
Fractals + 2LTP + GLCM	75.00	90.38			
Fractals + 3LTP	73.71	83.97			
GLCM + 8LTP	68.80	96.15			

Table 6 shows the outcomes of each group observed from different performance metrics. The results obtained from other performance measures also indicate the robustness of the solution for cancer detection.

Tabl	e 6.	E	xperiment	1—Fina	l result	s using	various	performance	e measures.
			1					1	

Group of Features	F1 (%)	ACC (%)	SENS (%)	SPEC (%)
Fractals + Wavelets + 8 LTP	94.87	94.61	97.43	93.10
Fractals + 8 LTP	88.46	89.74	91.02	93.10
GLCM	94.35	94.61	94.61	94.87
Fractals + 2 LTP + GLCM	92.16	92.69	92.69	90.38
Fracals + 3 LTP	95.38	95.51	95.51	97.43
GLCM + 8 LTP	95.89	96.15	96.15	97.43

7.2. Experiment 2

This experiment also used the DMR-IR database as input for the computation of features; however, the regions of interest in this experiment were slightly altered, corresponding only to the breast area, disregarding the possibility of node sentinel observation.

Table 7 illustrates the results of Experiment 2 considering the AUC on model selections and for a final classification of cancer or no cancer. It shows that the proposed methodology remains solid when using the features computed from slightly altered ROIs and with a final reduced range. It is possible to observe that the models generated in the final stage provide good AUC indices, surpassing the indices reached in the intermediate stage of the solution. The results reach good percentages of classification, reaching 94.87% of AUC. In comparison with the ROIs of Experiment 1 (where the armpit region is also considered), the results for some groups of features are slightly inferior; however, they remain satisfying.

	AUC (%)				
Group of Features	Intermediary (Model Selection)	Final (Feature Selection)			
Fractals + Wavelets + 8 LTP	81.41	91.02			
Fractals + 8LTP	81.41	92.94			
GLCM	83.97	94.87			
Fractals + 2LTP + GLCM	82.05	89.10			
Fractals + 3LTP	78.48	87.82			
GLCM + 8LTP	79.48	92.30			

Table 7. AUC for Experiment 2: Results on model selection and on cancer detection.

The results observed from other performance measures (all of them greater than 86%) also indicate that most patients were adequately diagnosed for the presence or absence of breast cancer, as presented in Table 8.

Table 8. Experiment 2—Final results using various performances.	

	F1 (%)	ACC (%)	SENS (%)	SPEC (%)
Fractals + Wavelets + 8 LTP	90.87	91.15	91.15	94.23
Fractals + 8 LTP	92.30	92.69	92.69	97.43
GLCM	94.35	94.61	94.61	94.87
Fractals + 2 LTP + GLCM	88.46	88.84	88.84	91.02
Fracals + 3 LTP	86.15	86.66	86.66	97.43
GLCM + 8 LTP	91.28	91.66	91.66	92.30

7.3. Experiments 3 and 4

Experiments 3 and 4 use a set of images from the UFPE database. The alteration of the database is interesting because it makes the evaluation of how the methodology behaves in other acquisition protocols and environmental conditions possible, thus analyzing if the results are biased by the used data. The used ROIs (segmented region) of these data also follow two different approaches corresponding to the breast and armpit areas (Experiment 3) and only the breast area (Experiment 4). Tables 9 and 10 illustrate these results.

Concerning Experiment 3, we can also observe that the diagnoses presented by the classifier combinations proposed in this work reach 97.91% of AUC for final classification, improving the intermediate step of the solution a lot. Similarly, the results of Experiment 4 also behave assertively, evolving between the intermediate and final stages. In both experiments, the diagnostics provided by the solution showed high rates of AUC. Moreover, it is also possible to observe that the other performance metrics also show similar results, with rates above 97%, as seen in Table 11.

Table 9. AUC for Experiment 3: Results on model selection and on cancer detection.

	AUC (%)				
Group of Features	Intermediary (Model Selection)	Final (Feature Selection)			
Fractals + Wavelets + 8 LTP	67.70	85.41			
Fractals + 8LTP	67.70	84.37			
GLCM	80.20	88.54			
Fractals + 2LTP + GLCM	83.85	97.91			
Fractals + 3LTP	62.50	85.41			
GLCM + 8LTP	82.29	97.97			

	AUC (%)				
Group of Features	Intermediary	Final			
	(Model Selection)	(Feature Selection)			
Fractals + Wavelets + 8 LTP	73.43	90.65			
Fractals + 8LTP	69.99	90.10			
GLCM	80.72	92.70			
Fractals + 2LTP + GLCM	76.64	89.06			
Fractals + 3LTP	83.85	87.50			
GLCM + 8LTP	80.20	86.97			

Table 10. AUC for Experiment 4: Results on model selection and on cancer detection.

Table 11. Experiments 3 and 4 evaluated by other pe	erformance indices
------------------------------------------------------------	--------------------

	F1		ACC		SENS		SPEC	
	#3	#4	#3	#4	#3	#4	#3	#4
Fractais + Wavelets + 8 LTP Fractais + 8 LTP GLCM Fractais + 2 LTP + GLCM Fractais + 3 LTP GLCM + 8 LTP	84.97 84.55 90.98 97.29 86.78 97.29	90.5 91.07 91.87 89.01 85.71 85.77	86.77 87.18 91.77 97.18 88.54 97.18	91.35 91.87 92.6 90.10 85.72 86.04	86.77 87.18 91.77 97.18 88.54 97.18	91.35 91.87 92.6 90.1 85.72 86.04	76.04 73.95 85.41 94.79 81.25 94.70	85.41 87.5 85.41 86.45 81.25 79.16

7.4. Considerations about the Computational Performances

Another contribution of the proposed methodology is the reduction of the processing amount. The selection of features performed substantially reduces the number of features required for diagnosis for all four of the experiments, as shown in Table 12. In practical terms, for future patients to be evaluated, the computation feature process and then the cancer detection will be minimized since we already know which features can better describe the problem.

In terms of performance, it is interesting that the approach proposed in this work provides substantial savings in terms of the number of trained models compared to the exhaustive search. Considering the entire set of possible parameters to be used by the classifier, an exhaustive search would train more than half a million models (precisely 540,560) for each group of features in each one of the experiments in this work.

On using the approach proposed in this work, the model selection stage requires only 0.77% of the total models performed by the exhaustive search, training a total of only 4200 models (140 individuals \times 30 generations). Considering the whole solution proposed in this work (model and feature selection), 18,200 models were trained for each group of features (140 individuals \times 30 generations for the model selection + 140 individuals \times 100 generations for the feature selections). This number represents only 3.37% of all the executions that the exhaustive search would carry out only in selecting models, without considering the feature selection process (crucial for a good classification), on which it would depend on another approach.

Table 12. Reduction in the number of features.

	Initial	al N°. of Features Selected				
	Features	#1	#2	#3	#4	
Fractais + Wavelets + 8 LTP	384	18	7	9	45	
Fractais + 8 LTP	48	16	23	14	17	
GLCM	48	17	17	15	15	
Fractais + 2 LTP + GLCM	60	10	7	23	9	
Fractais + 3 LTP	18	8	6	7	7	
GLCM + 8 LT	148	42	33	22	63	

7.5. Genetic Algorithm vs. Percentile Techniques for Feature Selection

We also compared the result obtained in this study with another technique for feature selection. In this analysis, we use the results from the model selection and, instead of improving them through the selection of features proposed in previous section, we now use the percentile as an approach for such a selection. The percentile is based on univariate statistical tests on the dataset and is also used to select features in works in the literature [57,58].

This comparison is presented in Table 13 and allows us to observe that our genetic proposal leads to better results in practically all the groups. In these cases, considering the six groups of features explored and the four experiments, the results are different in only two of them—even though, in one of them, it can be said that the results are highly similar: 87.5% vs. 86.97%, in the "GLCM + 8LTP" group, on Experiment #4.

	Experiments							
	#1		#2		#3		#4	
	Perc.	Gene.	Perc.	Gene.	Perc.	Gene.	Perc.	Gene.
Fractais + Wavelets + 8 LTP	70.83	94.87	70.83	91.02	79.16	85.41	90.62	90.65
Fractais + 8 LTP	69.44	96.15	75.00	92.94	79.16	84.37	93.75	90.10
GLCM	81.94	94.87	66.66	94.87	75.00	88.54	90.62	92.70
Fractais + 2 LTP + GLCM	63.88	90.38	50.00	89.10	84.37	97.91	88.54	89.06
Fractais + 3 LTP	63.88	83.97	75.00	87.82	82.29	85.41	82.29	87.50
GLCM + 8 LTP	81.94	96.15	65.27	92.30	90.62	97.91	87.50	86.97

Table 13. Comparing Genetic and Percentile techniques for feature selection.

7.6. Comparison with Related Work

In Section 2, we presented several studies focused on the classification of thermal images involving breast cancer. We can observe that the analyzed studies use combinations of features and classifiers associated with different strategies for screening and diagnosing breast cancer. Moreover, some studies employ complex machine learning techniques like deep learning. We compare our results to those studies and summarize such comparisons in Table 14.

Table 14. Proposed method comparison with related work.

Author	F1 (%)	ACC (%)	SENS (%)	SPEC (%)	AUC (%)
Arora et al. [11]	Not informed	Not informed	97.0	27.0	Not informed
Wishart et al. [12]	Not informed	Not informed	78.0	75	Not informed
Hossein et al. [13]	Not informed	70.0	50.0	75.0	Not informed
Krawczyk et al. [14]	Not informed	91.09	80.64	94.82	Not informed
Gerasimova et al. [16,17]	Not informed	Not informed	76.0	86.0	Not informed
Lashkari et al. [18]	68.56	87.42	68.56	92.14	80.35
Raghavendra et al. [19]	Not informed	98.0	96.66	100.0	98.0
Santana et al. [20]	Not informed	95.56	Not informed	Not informed	Not informed
Santana et al. [21]	Not informed	73.38	78.0	88.0	Not informed
Baffa and Lattari [23]	98.0	98.0	97.0	100.0	Not informed
Fernández-Ovies et al. [24]	Not informed	100.0	Not informed	Not informed	Not informed
Tello-Mijares et al. [25]	100.0	100.0	100.0	100.0	100.0
Sánchez-Ruiz et al. [28]	Not informed	98.33	96.67	100.0	Not informed
Mishra and Rath [29]	96.66	95.45	99.17	88.07	Not informed
Silva et al. [27]	Not informed	100.0	100.0	100.0	100.0
Silva et al. [22]	Not informed	87.96	Not informed	Not informed	Not informed
Proposed methodology	97.29	97.18	97.18	94.79	97.91

We consider the measures of greatest values between the screening and diagnosis phases.

Among related work, we observe that Arora et al. [11], Wishart et al. [12], Hossein et al. [13], Krawczyk et al. [14], Gerasimova et al. [16,17], Raghavendra et al. [19] Santana et al. [20,21], Fernández-Ovies et al. [24], Sánchez-Ruiz et al. [28], and Silva et al. [22,27] do not use the F1-score to assess the classifiers' performance. On the other hand, only Baffa and Lattari [23] and Tello-Mijares et al. [25] present an F1-score higher than ours, but both use CNN. However, it is worth highlighting that our work diagnoses cancer, while Tello-Mijares et al. [25] do screening. In other words, our work identifies cancer cases (diagnosis task), while Tello-Mijares et al. [25] only identify the presence of abnormalities in the breast (screening task). Furthermore, in general, our performance measures are very close to the performance obtained by Baffa and Lattari [23], except in values of SPEC.

Regarding the ACC measure, Krawczyk et al. [14], Raghavendra et al. [19], Santana et al. [20], Baffa and Lattari [23], Fernández-Ovies et al. [24], Tello-Mijares et al. [25], Sánchez-Ruiz et al. [28], Mishra and Rath [29], and Silva et al. [27] achieve ACC values greater than 90%. Analyzing Santana et al. [20], Fernández-Ovies et al. [24], and Silva et al. [22], we realized that they present ACC equal to 95.56%, 100.0%, and 87.96%, but they use an unbalanced dataset, and the ACC is not a good measure of performance in this case [39].

Examining Krawczyk et al. [14], we observed that SENS is considerably lower than ACC and SPEC, meaning that the study fails to diagnose positive outcomes for patients with breast cancer. Comparing our proposal with the study of Raghavendra et al. [19], we noticed that our results are very close to their performance, and we presented a higher score only in SENS. This result suggests that HOG features can provide a better description of thermal image features.

Arora et al. [11], Wishard et al. [12], Hossein et al. [13], and Gerasimova et al. [16,17] present the worst SPEC measures. Furthermore, comparing our work with Lashkari et al. [18], we realized that our work is robust and presents better results in all the measures analyzed, even without considering the entire amount of features and classifiers analyzed by them. Similarly, Santana et al. [21] present a lower performance compared to ours. Sánchez-Ruiz et al. [28] presented better performance compared to our work, except in the SENS measure, while Mishra and Rath [29] had the opposite response. These results are due to the choice of ANN and Random Forest classifiers.

The results obtained demonstrate that the approach proposed achieves performance very close to deep learning-based approaches, and this work presents a significant contribution to research on diagnosing breast cancer.

8. Conclusions

This work introduced a methodology for diagnosing breast cancer based on thermal imaging. For this, two ensemble strategies were proposed. The first one carried out a model selection using a technique called Bucket of Models. To do this, we combined the Genetic Algorithm and SVM to evaluate model results, perform a ranking, and select the best ones in a few generations (30 generations and 140 individuals per population).

The second ensemble method proposed in this work performed feature selection using another Genetic Algorithm, which also worked in conjunction with the SVM. However, the method aimed to obtain the smallest set of features that best classifies the data concerning breast cancer diagnosis at this stage. This step used as input the models best adapted to the problem, resulting from the model selection step, and performed the features selection over 100 generations of 140 individuals each. This task maximized the correctness of the classification since the reduction in the number of features can improve the classification results. Furthermore, we minimized the most demanding task and maximized fine-tuning by choosing the most significant features for the breast cancer classification [37].

We used breast thermal images from two datasets, DMR-IR and UFPE. These images were segmented following different approaches (with and without armpits) to compose four experiments. Armpits host some lymph nodes that have been used to evaluate prognosis results (this technique is called lymph node sentinel). Each experiment evaluates the behavior of the proposed methodology for detecting breast cancer from different images. Six groups of features were computed for the images of each experiment. These features summarize different aspects found in thermographies.

The variety of datasets, segmentation methods, and computed features allow us to assess the robustness of the methodology under different perspectives. In this way, it is possible to observe how much the proposed approaches can learn from the input data and find the sets of models and features that best adapt to the classification problem.

The results achieved by our Genetic Algorithms sequential combination were solid and satisfactory in all experiments, reaching rates of up to 97.91% AUC and 97.18% ACC in the detection of breast cancer. Other indices for quantification of the results also confirm these highest rates for the breast cancer detection process. In comparison with the literature, our work is also powerful and with competitive results on diagnosing breast cancer.

In addition, our proposal optimizes the model training process by more than 96%, compared to the exhaustive search, since the approach does not waste time exploring sets

of models and features that do not contribute to the classification process. Through the results obtained by our methodology, it was also possible to observe that it substantially reduces the final set of features that best add value to the classification process—thus minimizing the number of data that need to be preprocessed for classification.

The proposed methodology presented a solid contribution for diagnosing breast cancer using a Genetic Algorithm and Support Vector Machines classifier. This work, however, is subject to limitations that we intend to handle in the next stages of the study, such as: (1) Increase the number of exams used; (2) Investigate the feasibility of using strategies to generate synthetic data; (3) Explore the use of thermographies captured from other angles (e.g., laterals at 45° and 90°); (4) Include data of different breast diagnoses, such as benign tumors and cysts; (5) Explore the applicability of the methodology in more breast thermographic datasets; and (6) Investigate the performance of the methodology under other base classifiers.

Additionally, for the next steps, an evolution of the methodology may assess the difference between the right and left breasts in the anomaly detection process. Finally, the proposed classifier combination system can also be evaluated using datasets, computed from other types of examination, to assess the methodology's applicability for other open problems.

Author Contributions: Conceptualization: R.R. and A.C.; Data curation: R.R., A.S.A., P.M. and A.C.; Formal analysis: R.R. and A.C.; Investigation: R.R.; Methodology: R.R., D.M.-S. and A.C.; Resources: L.S., A.S.A. and P.M.; Supervision: D.M.-S. and A.C.; Validation: R.R., L.S., A.S.A., D.M.-S. and A.C.; Visualization: A.S.A., R.R.; Writing: R.R., L.S., A.S.A., P.M., D.M.-S. and A.C. All authors have read and agreed to the published version of the manuscript.

Funding: This research received funding from CAPES-PRINT (project: 88881.310304/2018-01), FAPERJ (project: E-26/010.002431/2019), CNPq/INCT (project: 307329/2016-0) and CAPES PROEX (project: 23038.000949/2018-15).

Institutional Review Board Statement: The study was conducted according to the guidelines of the Declaration of Helsinki. The thermographic datasets were acquired attending the Research Ethics Committee of the belonging organization. The first one, DMR-IR, is registered under the CAAE number: 01042812.0.0000.5243 at the Brazilian Ministry of Health platform. The second of them, the UFPE database, is also registered in the Brazilian Health Ministry under CEP/CCS/UFPE 279/05.

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

Data Availability Statement: All data used in this research were obtained from two breast thermographic datasets. The first one, the DMR-IR, is completely available at http://visual.ic.uff.br/dmi (accessed on 1 July 2021). The second oh them is a private thermal image database of the Federal University of Pernambuco (UFPE).

Acknowledgments: The authors are thankful to the patients who consented to the use of their data for research. R.R., L.S., A.S.A. and P.M. gratefully acknowledge the support from CAPES in their graduation courses. A.C. and D.M. thank CAPES PRINT, FAPERJ, INCT-MACC, and CNPq.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Sung, H.; Ferlay, J.; Siegel, R.L.; Laversanne, M.; Soerjomataram, I.; Jemal, A.; Bray, F. Global Cancer Statistics 2020: GLOBOCAN Estimates of Incidence and Mortality Worldwide for 36 Cancers in 185 Countries. CA Cancer J. Clin. 2021, 71, 209–249. [CrossRef]
- 2. International Agency for Research on Cancer (IARC). *Cancer Tomorrow;* International Agency for Research on Cancer (IARC): Lyon, France, 2021.
- Borchartt, T.B.; Conci, A.; Lima, R.C.; Resmini, R.; Sanchez, A. Breast thermography from an image processing viewpoint: A survey. *Signal Process.* 2013, 93, 2785–2803. [CrossRef]
- 4. Doi, K. Computer-Aided Diagnosis in Medical Imaging: Historical Review, Current Status and Future Potential. *Comput. Med. Imaging Graph. Off. J. Comput. Med. Imaging Soc.* 2007, *31*, 198–211. [CrossRef]
- Arabi, P.M.; Muttan, S.; Suji, R.J. Image enhancement for detection of early breast carcinoma by external irradiation. In Proceedings of the 2010 Second International Conference on Computing, Communication and Networking Technologies, Bangkok, Thailand, 23–25 April 2010; pp. 1–9. [CrossRef]

- 6. Damiao, C.; Gonzalez, J.R.; Moran, M.B.H.; Fontes, C.P.; Balarine, G.; Cruz Filho, R.; Conci, A. On the possibility of using temperature to aid in thyroid nodule investigation. *Sci. Rep.* **2020**, 2045–2322. [CrossRef]
- Vogler, W.; Powell, R. Clinical Evaluation of Thermography and Heptyl Aldehyde in Breast Cancer Detection. *Cancer Res.* 1959, 19, 207–209.
- 8. Strax, P. Thermography in Mass Screening for Breast Cancer. Analls N. Y. Sci. 1964, 121, 282. [CrossRef]
- 9. Connell, J.; Ruzicka, F.; Grossi, C.; Osborne, A.; Conte, A. Thermography in the Detection of Breast Cancer. *Cancer* **1966**, *19*, 83–88. [CrossRef]
- 10. Al Husaini, M.A.S.; Habaebi, M.H.; Hameed, S.A.; Islam, M.R.; Gunawan, T.S. A Systematic Review of Breast Cancer Detection Using Thermography and Neural Networks. *IEEE Access* 2020, 208922–208937. [CrossRef]
- 11. Arora, N.; Martins, D.; Ruggerio, D.; Tousimis, E.; Swistel, A.J.; Osborne, M.P.; Simmons, R.M. Effectiveness of a noninvasive digital infrared thermal imaging system in the detection of breast cancer. *Am. J. Surg.* **2008**, *196*, 523–526. [CrossRef]
- Wishart, G.; Campisi, M.; Boswell, M.; Chapman, D.; Shackleton, V.; Iddles, S.; Hallett, A.; Britton, P. The accuracy of digital infrared imaging for breast cancer detection in women undergoing breast biopsy. *Eur. J. Surg. Oncol. (EJSO)* 2010, 36, 535–540. [CrossRef]
- 13. Hossein, G.; Javad, H.; Maryam, H.; Kazem, H. Diagnosis of Breast Cancer Using a Combination of Genetic Algorithm and Artificial Neural Network in Infrared Thermal Imaging. *Iran. J. Med. Phys.* **2012**, *9*, 265–274.
- Krawczyk, B.; Schaefer, G.; Wozniak, M. Breast thermogram analysis using a cost-sensitive multiple classifier system. In Proceedings of the 2012 IEEE-EMBS International Conference on Biomedical and Health Informatics, Hong Kong, China, 5–7 January 2012; pp. 507–510. [CrossRef]
- 15. Opitz, D.; Maclin, R. An empirical evaluation of bagging and boosting for artificial neural networks. In Proceedings of the 1997 IEEE International Conference on Neural Networks (ICNN 97), Houston, TX, USA, 12 June 1997; Volume 1–4. pp. 1401–1405.
- 16. Gerasimova, E.; Audit, B.; Roux, S.; Khalil, A.; Argoul, F.; Naimark, O.; Arneodo, A. Multifractal analysis of dynamic infrared imaging of breast cancer. *Europhys. Lett. (EPL)* **2014**, *104*, 68001. [CrossRef]
- 17. Gerasimova, E.; Audit, B.; Roux, S.G.; Khalil, A.; Gileva, O.; Argoul, F.; Naimark, O.; Arneodo, A. Wavelet-based multifractal analysis of dynamic infrared thermograms to assist in early breast cancer diagnosis. *Front. Physiol.* **2014**, *5*. [CrossRef] [PubMed]
- 18. Lashkari, A.; Pak, F.; Firouzmand, M. Full Intelligent Cancer Classification of Thermal Breast Images to Assist Physician in Clinical Diagnostic Applications. J. Med. Signals Sens. 2016, 6, 12–24. [CrossRef] [PubMed]
- Raghavendra, U.; Acharya, U.R.; Ng, E.Y.K.; Tan, J.H.; Gudigar, A. An integrated index for breast cancer identification using histogram of oriented gradient and kernel locality preserving projection features extracted from thermograms. *Quant. InfraRed Thermogr. J.* 2016, 13, 195–209. [CrossRef]
- Santana, M.; Pereira, J.M.; Lima, N.; Sousa, F.; Lima, R.; Dos Santos, W. Classificação de Lesões em Imagens Frontais de Termografia de Mama a Partir de Sistema Inteligente de Suporte ao Diagnóstico. In *I Simpósio de Inovação em Engenharia Biomédica* (SABIO 2017); BioTech Consultoria: Recife, Brazil, 2017; pp. 1–5.
- Santana, M.; Pereira, J.; Silva, F.; Lima, N.; Sousa, F.; Arruda, G.; Lima, R.; Silva, W.; Santos, W. Breast cancer diagnosis based on mammary thermography and extreme learning machines. *Int. J. Artif. Intell. Mach. Learn. (IJAIML)* 2018, 34, 45–53. [CrossRef]
- Silva, A.; Santana, M.; Lima, C.; Andrade, J.; Souza, T.; Almeida, M.; Silva, W.; Lima, R.; Santos, W. Features Selection Study for Breast Cancer Diagnosis Using Thermographic Images, Genetic Algorithms, and Particle Swarm Optimization. *Int. J. Artif. Intell. Mach. Learn.* (IJAIML) 2021, 11, 1–18. [CrossRef]
- 23. Baffa, M.F.O.; Lattari, L.G. Convolutional Neural Networks for Static and Dynamic Breast Infrared Imaging Classification. In Proceedings of the 2018 31st SIBGRAPI Conference on Graphics, Patterns and Images (SIBGRAPI), Parana, Brazil, 29 October–1 November 2018; pp. 174–181. [CrossRef]
- Fernández-Ovies, F.; Alférez, S.; DeAndrés-Galiana, E.J.; Cernea, A.; Fernández-Muñiz, Z.; Fernández-Martínez, J.L. Detection of Breast Cancer Using Infrared Thermography and Deep Neural Networks. In *Bioinformatics and Biomedical Engineering*; Rojas, I., Valenzuela, O., Rojas, F., Ortuño, F., Eds.; Springer International Publishing: Cham, Switzerland, 2019; pp. 514–523. [CrossRef]
- 25. Tello-Mijares, S.; Woo, F.; Flores, F. Breast Cancer Identification via Thermography Image Segmentation with a Gradient Vector Flow and a Convolutional Neural Network. *J. Healthc. Eng.* **2019**, 2019, 9807619. [CrossRef]
- 26. Silva, L.; Muchaluat-Saade, D.; Sequeiros, G.; Silva, A.; Paiva, A.; Bravo, R.; Conci, A. A new database for breast research with infrared image. *J. Med Imaging Health Inform.* **2014**, *4*, 92–100. [CrossRef]
- 27. Silva, T.A.E.; Silva, L.F.; Muchaluat-Saade, D.C.; Conci, A. A Computational Method to Assist the Diagnosis of Breast Disease Using Dynamic Thermography. *Sensors* 2020, *20*, 3866. [CrossRef]
- 28. Sánchez-Ruiz, D.; Olmos-Pineda, I.; Olvera-López, J.A. Automatic region of interest segmentation for breast thermogram image classification. *Pattern Recognit. Lett.* **2020**, *135*, 72–81. [CrossRef]
- 29. Mishra, V.; Rath, S.K. Detection of breast cancer tumours based on feature reduction and classification of thermograms. *Quant. InfraRed Thermogr. J.* 2020, 1–14. [CrossRef]
- 30. Kuncheva, L.I.; Whitaker, C.J. Measures of diversity in classifier ensembles and their relationship with the ensemble accuracy. *Mach. Learn.* **2003**, *51*, 181–207. [CrossRef]
- Kuncheva, L.; Bezdek, J.C.; Sutton, M.A. On combining multiple classifiers by fuzzy templates. In Proceedings of the 1998 Conference of the North American Fuzzy Information Processing Society-NAFIPS (Cat. No. 98TH8353), Pensacola Beach, FL, USA, 20–21 August 1998; pp. 193–197.

- 32. Kuncheva, L.I. Combining Pattern Classifiers: Methods and Algorithms; John Wiley & Sons: Hoboken, NJ, USA, 2014.
- 33. Tan, P.N.; Steinbach, M.; Kumar, V. Introduction to Data Mining; Pearson Education India: London, UK, 2016.
- 34. Da, Gama, J.M.P. Combining Classification Algorithms. Ph.D. Thesis, University of Porto, Porto, Portugal, 1999.
- 35. Burger, E.M.; Moura, S.J. Gated ensemble learning method for demand-side electricity load forecasting. *Energy Build.* 2015, 109, 23–34. [CrossRef]
- 36. Madasamy, K.; Ramaswami, M. Data imbalance and classifiers: Impact and solutions from a big data perspective. *Int. J. Comput. Intell. Res.* **2017**, *13*, 2267–2281.
- 37. Kuncheva, L.I.; Matthews, C.E.; Arnaiz-González, Á.; Rodríguez, J.J. Feature Selection from High-Dimensional Data with Very Low Sample Size: A Cautionary Tale. *arXiv* 2020, arXiv:2008.12025.
- 38. Resmini, R.; Faria da Silva, L.; Medeiros, P.R.; Araujo, A.S.; Muchaluat-Saade, D.C.; Conci, A. A Hybrid Methodology for Breast Screening and Cancer Diagnosis Using Thermography. *Comput. Biol. Med.* **2021**, 104553. [CrossRef]
- 39. Batista, G.E.A.P.A.; Prati, R.C.; Monard, M.C. A Study of the Behavior of Several Methods for Balancing Machine Learning Training Data. *ACM SIGKDD Explor. Newsl.* **2004**, *6*, 20–29. [CrossRef]
- 40. Kasabov, N.K. Foundations of Neural Networks, Fuzzy Systems, and Knowledge Engineering; MIT Press: Cambridge, MA, USA 1996.
- 41. Dzeroski, S.; Zenko, B. Is combining classifiers better than selecting the best one? Mach. Learn. 2004, 54, 255–273. [CrossRef]
- 42. Witten, I.H.; Frank, E.; Hall, M.A.; Pal, C.J. *Practical Machine Learning Tools and Techniques*; Morgan Kaufmann: Burlington, MA, USA, 2005; p. 578.
- 43. Whitley, D. A genetic algorithm tutorial. Stat. Comput. 1994, 4, 65–85. [CrossRef]
- 44. Holland, J.H. Adaptation in Natural and Artificial Systems: An Introductory Analysis with Applications to Biology, Control, and Artificial Intelligence; MIT Press: Cambridge, MA, USA, 1992.
- dos Santos Araujo, A.; Resmini, R.; Moran, M.B.H.; de Sousa Issa, M.H.; Conci, A. Computer Techniques for Detection of Breast Cancer and Follow Up Neoadjuvant Treatment: Using Infrared Examinations. In *Biomedical Computing for Breast Cancer Detection* and Diagnosis; IGI Global: Hershey, PA, USA, 2021; pp. 72–106.
- 46. Haralick, R.M. Statistical and structural approaches to texture. Proc. IEEE 1979, 67, 786–804. [CrossRef]
- 47. Tan, X.; Triggs, B. Enhanced local texture feature sets for face recognition under difficult lighting conditions. *IEEE Trans. Image Process.* **2010**, *19*, 1635–1650. [CrossRef] [PubMed]
- 48. Vonesch, C.; Blu, T.; Unser, M. Generalized Daubechies Wavelet Families. *IEEE Trans. Signal Process.* 2007, 55, 4415–4429. [CrossRef]
- 49. Higuchi, T. Approach to an irregular time series on the basis of the fractal theory. *Phys. D Nonlinear Phenom.* **1988**, *31*, 277–283. [CrossRef]
- Petrosian, A. Kolmogorov complexity of finite sequences and recognition of different preictal EEG patterns. In Proceedings of the Eighth IEEE Symposium on Computer-Based Medical Systems, Lubbock, TX, USA, 9–10 June 1995; pp. 212–217. [CrossRef]
- 51. Hurst, H.E. Long-term storage capacity of reservoirs. *Trans. Am. Soc. Civ. Eng.* **1951**, *116*, 770–799. [CrossRef]
- 52. Chang, C.C.; Lin, C.J. LIBSVM: A library for support vector machines. *ACM Trans. Intell. Syst. Technol. (TIST)* 2011, 2, 1–27. [CrossRef]
- 53. Huang, M.W.; Chen, C.W.; Lin, W.C.; Ke, S.W.; Tsai, C.F. SVM and SVM ensembles in breast cancer prediction. *PLOS ONE* 2017, 12, e0161501. [CrossRef] [PubMed]
- 54. Akay, M.F. Support vector machines combined with feature selection for breast cancer diagnosis. *Expert Syst. Appl.* 2009, *36*, 3240–3247. [CrossRef]
- 55. Pedregosa, F.; Varoquaux, G.; Gramfort, A.; Michel, V.; Thirion, B.; Grisel, O.; Blondel, M.; Prettenhofer, P.; Weiss, R.; Dubourg, V.; et al. Scikit-learn: Machine learning in Python. *J. Mach. Learn. Res.* **2011**, *12*, 2825–2830.
- Van der Walt, S.; Schönberger, J.L.; Nunez-Iglesias, J.; Boulogne, F.; Warner, J.D.; Yager, N.; Gouillart, E.; Yu, T. scikit-image: Image processing in Python. *PeerJ* 2014, 2, e453. [CrossRef]
- 57. Pham, T.C.; Luong, C.M.; Doucet, A.; Hoang, V.D.; Tran, D.P.; Le, D.H. Meta-analysis of computational methods for breast cancer classification. *Int. J. Intell. Inf. Database Syst.* 2020, *13*, 89–111. [CrossRef]
- Shao, C.; Paynabar, K.; Kim, T.H.; Jin, J.J.; Hu, S.J.; Spicer, J.P.; Wang, H.; Abell, J.A. Feature selection for manufacturing process monitoring using cross-validation. J. Manuf. Syst. 2013, 32, 550–555. [CrossRef]